REVIEW ARTICLE

The effect of family involvement in diabetes self-management education on glycemic control in patients with type 2 diabetes mellitus: a systematic review and meta-analysis

Akhmad Azmiardi^{1*}, Bhisma Murti², Ratih Puspita Febrinasari³, Didik Gunawan Tamtomo²

¹ Doctoral Program of Public Health, Universitas Sebelas Maret, Surakarta, Indonesia

² Doctor of Public Health Program, Universitas Sebelas Maret, Surakarta, Indonesia

³ Department of Pharmacology, Faculty of Medicine, Universitas Sebelas Maret Surakarta, Indonesia

Corresponding author: Akhmad Azmiardi Email: aazmiardi@gmail.com

Received: 10 February 2021 Revised: 31 March 2021 Accepted: 14 April 2021 Available online: September 2021

ABSTRACT

Diabetes self-management education (DSME) is an essential part of diabetes care. It is possible to deliver DSME in many forms and components. In this case, family support is known to be a critical component of social support in self-care and glycemic control. This study aimed to systematically review the effectiveness of family involvement in diabetes self-management education for glycemic control in patients with type 2 diabetes mellitus. The researchers searched PubMed, SpringerLink, Science Direct, Google Scholar, and EBSCO for English and Indonesian articles published between 2000 and 2020 describing randomized controlled trials involving family components. The primary outcome of this review was HbA1c changes. The pooled standard mean differences (SMD) or effect size between intervention and control groups with 95% confidence interval (CI) were calculated using a random-effects model. Meanwhile, heterogeneity of HbA1c findings was assessed with Cochran's Q and I2. Seventeen randomized controlled trials with a total of 2644 participants were obtained from a comprehensive search procedure. The intervention was effective in reducing HbA1c (SMD= -0.31 95%CI= -0.46 to -0.16;) than the control group, and it was statistically significant (p < .001). Due to heterogeneity (I2 = 71%), a meta-analysis of the random-effects model was employed. Intervention program with baseline HbA1c < 9% (SMD= -0.55; 95%CI= -1.05 to -0.05; p=.030), duration of intervention ≥ 6 month (SMD= -0.26; 95%CI= -0.37 to -0.15; p<.001), intervention delivery mode of combination type (SMD= -0.73; 95%CI= -1.42 to -0.03; p=.040), contact hour ≥ 10 (SMD= -0.37; 95%CI= -0.59 to -0.14; p=.001), and frequency of contact (SMD= -0.75; 95%CI= -1.39 to -0.12; p=.020) were effective to reduce HbA1c level. According to the Grading of Recommendations Assessment, Development, and Evaluation (GRADE) method, the quality of evidence was classified as moderate. Family involvement in the DSME program was effective in improving glycemic control in patients with type 2 diabetes. Further research on baseline HbA1c, delivery mode, contact hours, and intervention frequency may provide useful information to determine the most effective treatment.

Key words: family involvement, randomized controlled trial, type 2 diabetes, meta-analysis

INTRODUCTION

Type 2 diabetes mellitus is recognized as a global public health issue with a substantial effect on the cost of living and human health. In many areas of the world, rapid economic growth and urbanization have contributed to a growing burden of diabetes¹.

People with diabetes are at greater risk of experiencing a number of severe life-threatening health complications, renal disease, paralysis, heart attacks, strokes, and lower-limb amputations, resulting in a high socio-economic burden and quality of life, leading to higher medical care expenses, lower quality of life, and increased mortality². Besides, type 2 diabetes tends to escalate the prevalence, severity, and leading cause of human suffering and death. Approximately 462 million people had type 2 diabetes in 2017, equivalent to 6.28% of the world's population. As many as 4.4% of them are aged 15-49 years, 15% are aged 50-69, and 22% are more than 70 years old³.

If no effective preventive measures are adopted, the International Diabetes Federation (IDF) has projected a rise of 693 million people with diabetes by 2045⁴. For people with diabetes, there is a marginally higher chance of mortality than in the population without diabetes. Body Mass Index (BMI), systolic blood pressure, total cholesterol and HDL, increased intake of unhealthy foods, sedentary lifestyle, and smoke culminated in elevated fasting plasma glucose. These factors, separately, show a strong association with mortality rates⁵.

Furthermore, diabetes is a complex chronic disease that requires continuing medical care. Risk prevention strategies by education and self-management are essential to prevent acute complications and reduce the risk of long-term complications⁶. Self-management of people with type 2 diabetes can be accomplished by improving understanding of risk factors, symptoms, nutrition, physical exercise, glycemic monitoring, and insulin prescription and utilization. The cornerstone of diabetes is to improve the patient's diet by effective treatment recommendations to minimize diseaserelated mortality and morbidity⁷.

Diabetes self-management education (DSME) is a crucial element in caring for both people with diabetes and those at risk of developing the condition. Besides, diabetes complications need to be avoided or postponed in the effective management of people with diabetes⁸. The initial stage of DSME is to promote the awareness, expertise, and ability of pre-diabetes and diabetes self-care practitioners to control their diabetes⁹. Regarding this, the American Diabetes Association (ADA) encourages self-management education and support at least yearly to avoid acute complications of diabetes mellitus and minimize the risk of long-term complications¹⁰. Previous studies have revealed that DSME is associated with increased diabetes awareness and decreased HbA1c¹¹⁻¹³.

The hemoglobin A1c test (glycated hemoglobin, glycosylated hemoglobin, HbA1c, or A1c) is used to determine an individual's glucose regulation level. The test indicates the average blood sugar level for the last 90 days or three months. HbA1c level greater than 6.5 percent is known to be uncontrolled glycemia¹⁴.

On the other hand, diabetes treatment can be relatively difficult for patients. They must attend multiple medical appointments per year, adhere to a number of different medications to manage their disease, engage in several forms of self-care, including home glucose screening and a healthy diet and lifestyle, and negotiate maintenance costs⁶. As a chronic disease, most treatments for this disease are performed at home and within the family. Thus, diabetes is often referred to as a family disorder, and its controls and demands influence all family members as such social support, especially family support, can be a significant part of effective diabetes and glycemic control¹⁵.

Further, family members are known to be a key source of social support for adults with diabetes and an effective and critical component in self-care and disease control¹⁶. It is also vital for these patients, their families, and the health care teams to be engaged in the treatment process to manage the disease. Both approaches from family and health care professionals are combined with the recommendations on diabetes and the determination of patients to adhere to a diet plan, exercise, medical therapy, support, and supervision from both.

Previous systematic review findings have demonstrated that familysupported DSME increases clinical outcomes for patients with untreated glycemia¹⁷. The author included 22 experiments conducted between 2008 and 2016. The research findings found that family support improved the clinical conditions of patients with untreated glycemia. Despite systematic reviews of family support in DMSE have been published, no metaanalyses have specifically analyzed the impact of family involvement in DSME with type 2 diabetes. Furthermore, no studies have examined and described the characteristics, such as baseline HbA1c, type of delivery, contact hours, or intervention frequency, that might yield better glycemic control outcomes for patients with type 2 diabetes. There is limited knowledge about the impact of such self-management intervention for patients with type 2 diabetes. Therefore, this systematic review and meta-analysis were carried out to identify the knowledge gaps and to make recommendations for primary research. This study aimed to assess the

effect of family involvement in DSME compared to usual treatment on glycemic control in DSME programs in type 2 diabetes patients. Several characteristics of family involvement in DSME were also investigated in subgroup analyses to identify who have better or worse glycemic control. The findings will promote the preparation of evidence-based interventions and help inform future studies.

MATERIALS AND METHODS

Design

This research applied a systematic review and meta-analysis. All similar studies, including topic and design studies, were gathered through a systematic review and meta-analysis, after which study findings were checked and reanalyzed. The Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) were used to establish and publish systematic review and meta-analysis¹⁸. Besides, a systematic review and meta-analysis approach was employed to comply with the principles of the Cochrane handbook for systematic reviews of interventions¹⁹.

Search strategy

This study was carried out by searching and selecting data from clinical trial results conducted across ethnicities, races, and countries worldwide. 2000-2020 was the selected study outcome period. Multiple journal databases, including PubMed, SpringerLink, Science Direct, Google Scholar, and EBSCO, were utilized to search for study results. Besides, medical subject (MeSH) terms and publications of the PICO framework (participants, intervention, comparison, and outcomes) were employed for the search strategy. The search keywords used were "type 2 diabetes" AND "self-management" AND "diabetes self-management education" AND "family support" AND "HbA1c" AND/OR "glycemic".

To find additional studies adequate for this study's purposes, the researchers reviewed the articles' reference lists. The researcher selected articles published in Indonesian and English. Meanwhile, other qualifying experiments were manually scanned for references from the chosen studies.

Inclusion and exclusion criteria

In this meta-analysis study, the inclusion criteria were that: (1) the study design was a randomized controlled trial (RCT); (2) the study subjects were people with type 2 diabetes mellitus; (3) a comparison group without any active DSME component was used, such as routine treatment that may consist of regular primary care, a waiting list, or limited intervention. Studies eligible for inclusion were also required to meet the DSME definition by the National Standards for Diabetes Self-Management Education and Support, (4) there was family participation involved in DSME interventions, (5) the intervention time was at least three months as HbA1c represents a mean glycemia of approximately three months, (6)improvements in Glycohemoglobin A1c were recorded, and (7) in this review, the papers chosen were in full text.

Moreover, Diabetes Self-Management Education (DSME) is the method to promote the awareness, knowledge, and skills required for self-care for diabetes ²⁰. This definition is not prescriptive but more process-oriented; DSME interventions should provide elements and practices to strengthen participants' awareness, expertise, and capacity to carry out self-management activities that can enhance glycemic control. Eligible for inclusion were DSME treatments delivered in any environment, by any method or provider, and for any period and contact hour.

Meanwhile, the exclusion criteria included that (1) abstracts, dissertations,

and conference papers were excluded because they included inadequate details. (2) Articles were excluded in the absence of randomized control trials. (3) Pilot studies, single cohort feasibility studies, or policy reviews were excluded. (4) Studies that did not involve type 2 diabetes mellitus as a primary disease were excluded, including type 1 diabetes, asthma, or arthritis (4). The study step was carried out with the Preferred Reporting Items for Systematic Review and Meta-Analysis (PRISMA)¹⁸. In Figure 1, the selection study flow chart is presented.

Study selection

A screening method and objective evaluation for the chosen papers were performed by two reviewers (AA and TH). Based on predefined inclusion and exclusion criteria, the reviewers obtained information from the abstracts and full texts. Meanwhile, both duplicate citations and documents from the same article found were deleted. The reviewers extracted information from the abstract into a structured table independently. To compare the findings obtained, the two reviewers read and graded the full-text articles. Through discussion and agreement, the differences in the resulting outcomes were resolved. Articles that qualify for a systematic analysis were included.

Data extraction

Selected articles were evaluated and extracted separately. The data were obtained from the selected articles, including the first author's name, publication year, the study's design, country, respondents' age, sample size, delivery program, contact hour, contact frequency, intervention duration, and HbA1c.

Furthermore, several subgroups analyses were carried out for HbA1c. The intervention duration was classified as <six months and \geq six months. The contact frequency was calculated based on the recorded intervention procedure and when it was available. The participants' HbA1c levels were at baseline, whereas the contact frequency was categorized into three levels: low (less than one contact per one month per patient), moderate (one or two contacts per one month per patient), and high (more than two contacts a month per patient). Contact hours were grouped as <10 and 10 hours, while the type of program delivery was measured for individual, group, or a combination of individual and group deliveries.

Bias and quality appraisal

The Cochrane Collaboration tool was utilized in the bias risk assessment, with the following categories: selection bias (random sequence generation and allocation concealment), performance bias (blinding participants and researchers to the intervention received by participants), detection bias (blinding of knowledge result assessment of what intervention a participant received), attrition bias. reporting bias, and other bias ¹⁹. This tool used the relevant criteria for classification as low, unclear, or high risk. As suggested, "low risk of bias" is advised if the consistency requirements for the least bias are completely reached; "unclear" if it is probable that a bias poses some concern about the results; "high risk of bias" if it is possible that a bias severely weakens confidence in the outcomes.

Determining the quality of the research obtained is crucial in choosing the studies to be included in the systematic review. Therefore, the researchers used the Grading of Recommendations. Assessment, Development, and Evaluation (GRADE). In the GRADE System, the assessment results can be categorized as high, medium, low, or very low. Research with a randomized controlled trial design is considered high quality, whereas

observational studies are of low quality. However, factors such as limitations in heterogeneity, indirectness, design, imprecision, and publication bias can reduce the assessment of the article's quality²¹. Thus, specifically, publication bias was assessed by funnel plot, Egger's test, and Begg's test. The funnel plot is a scatter plot based on the predicted effect size on the sample size. The funnel plots will be distributed symmetrically if there is no publishing bias. On the other hand, if the funnel is asymmetrical, publication bias is indicated ²². Furthermore, p-values of less than 0.05 in Begg's test and Egger's test were considered statistically significant if there was potential publication bias ^{23,24}.

Meta-analysis

Typically, statistical analysis is performed in Review Manager 5.2 and Stata version 16. In this study, the metaanalysis utilized the Review Manager (REVMAN) version 5.2 to determine the intervention's effect on the HbA1c. Data analysis used the mean and standard deviation of the results of measuring the HbA1c level before and after the intervention. Meta-analysis was also conducted to determine the differences in baseline HbA1c between the intervention and control groups. Besides, the mean HbA1c values for the intervention and control groups were presented as differences in standard mean, 95% CI, and heterogeneity $(I^2).$ Then. the data heterogeneity employed Cochran's Q and I^2 , with p-values <.05 for Cochran's Q values and $I^2 \ge 50\%$, suggesting substantial heterogeneity ²⁵.

The results were calculated and aggregated in random-effects metaanalysis. The observed random-effects meta-analysis varied across experiments due to variations in each study's treatment effects and heterogeneity in sampling²⁶. The meta-analysis findings were presented in the form of a forest plot for change in HbA1c. A forest plot is a diagram displaying each experiment's details in the meta-analysis and predictions of the total effects. The forest plot visually reveals the large variation (heterogeneity) between the outcomes. Apart from assessing the magnitude of variation in the forest plot, it also shows the strength of the relationship²⁷.

RESULTS

Study Characteristics

Figure 1 exhibits the PRISMA diagram in the article selection process.

The initial search process resulted in 991 articles obtained from PubMed, SpringerLink, Science Direct, Google Scholar, and EBSCO databases. After removing duplicate articles, 889 articles remained. After the abstract review process, 74 articles were selected for further full review. Then, there were 25 articles left for critical appraisal. In the end, 17 of the 25 articles were selected for a systematic review.



Figure 1 PRISMA flow chart

Table 1 shows the characteristics of the studies included in systematic reviews. Publication time ranged from 2000 to 2020. All articles were randomized controlled trials (RCTs). Articles varied in sample size, duration, and DSME interventions with family involvement. The number of respondents from all articles included in this systematic review totaled 2644 respondents. The respondents' mean age was 56.16 years (ranging from 61.3 to 49.1 years). The sample sizes were diverse from

17 to 286. All respondents considered patients with type 2 diabetes. The mean length of diabetes was 8.2 years (ranging from 3.8 to 16 years). DSME interventions with family involvement were different in setting, structure, content, type, duration, frequency, contact hours and providers. intervention The location was heterogenous from primary care offices, hospitals, community health centers, diabetes education centers, and patient homes. DSME interventions with family involvement were delivered by various experts, such as diabetes educators, nurse educators, nurse case managers, registered dieticians, pharmacists, and peer educators. All these interventions involved the family in their implementation.

The majority of the studies were carried out in the United States^{28–37},

Iran^{38,39}, and Thailand^{40,41}, followed by Ireland⁴², Taiwan ⁴³, and Brazil ⁴⁴. Six studies had baseline of $< 9^{29,31-33,39,40}$, and eleven studies had baseline HbA1c of \geq **Q**28,30,44,34–38,41–43 Four studies had intervention duration of <6 months^{29,32,39,40}. and 13 studies had an intervention duration of > 6 months^{28,30,42-44,31,33-38,41}. Five studies conducted individually^{33,37,41,42,45}; were carried out eight studies were in groups^{28,29,31,35,38-40,44}; four were done in combination with individuals and groups in the delivery of the DSME intervention 30,32,36,43. Six articles had contact hours of $< 10^{39-44}$, and 11 articles had contact hours of $> 10^{28,29,38,30-37}$. Four studies had a low contact frequency^{39–41,43}, four studies were moderate^{32,37,38,44}, and nine studies had a high contact frequency^{28–31,33–36,42}.

			р.:	Sam	ple (N)	Dia durati	betes ion (yrs)	Mea	n age	Definition	C hou	ontact 1r/frequ	Type of			Durati	Mean	HbA1c
Author	Year	Country	Design	IG	CG	IG	CG	IG	CG	of Family	c	ency ontact	deliver y	Family involvement in DSME intervention	Control	on	IG (Mean/SD)	CG (Mean/SD)
Brown ²⁸	2002	US	RCT	126	126	7.6	8.1	54.7	53.3	First degree relative	>10	>2	Group	DSME was provided by bilingual Mexican American nurses, dietitians, and community workers accompanied by a family member or friend. The intervention involved: (1) 3 months of	Usual care	12 month	10.89 (2.56)	11.64 (2.85)
														weekly instructional sessions on nutrition, self- monitoring of blood glucose, exercise, and other self-care topics; and (2) 6 months of biweekly				
														support group sessions to promote behavior changes. The approach was culturally competent in				
														terms of language, diet, social emphasis, family participation, and incorporation of cultural health beliefs				
Vincent ²⁹	2009	US	RCT	9	8	7,9	7,8	56.67	55.25	Friends and family	>10	>2	Group	The intervention delivered by physician or nurse consisted of eight weekly 2-hr group sessions, which included didactic content, cooking demonstrations, and group support (friends and family). Participants were encouraged to bring a support person to the sessions.	Usual care	3 month	6.14 (0.5)	6.84 (1,3)
Kang ⁴³	2010	Taiwan	RCT	33	34	3.8	4.4	55.3	51.7	Family members, parent, significant other, or additional important relative.	<10	1	Both	This intervention/ FPIC group consisted of three brief individual educational sessions, 2-day long group educational sessions, a monthly 25-30 min telephone discussion. All patients and family members also received diabetes handouts about diet, medication, physical activity and exercise, and eye and foot self-care at the first IES. Each education session lasted 20-40 min. Intervention delivered by physician, nurse, dietitian, pharmacist, physiotherapist, social worker	Usual care	6 month	7.9(1.36)	8.12(1.21)

Table 1 Summary of 17 randomized controlled trials family involvement in DSME on glycemic control in patients with type 2 diabetes included in the analysis

				Sam	ple (N)	Dia durati	betes on (yrs)	Mea	n age	Definition	Ca hou	ontact r/frequ	Type of		~	Durati	Mean	HbA1c
Author	Year	Country	Design	IG	CG	IG	CG	IG	CG	of Family	e co	ency ntact	deliver y	Family involvement in DSME intervention	Control	on	IG (Mean/SD)	CG (Mean/SD)
Keogh ⁴²	2011	Ireland	RCT	60	61	9.17	9.65	59.96	57.29	Family members in the same home as patients	<10	>2	Individ ual	The intervention consisted of 3 weekly sessions consisted of 2 sessions delivered by a health psychologist to the patient and a family member in the patient's home. The first 2 sessions lasted 45 minutes each. The third session involved a 10- to 15-minute follow-up telephone call.	Usual diabetes care	6 month	8.41 (0.99)	8.80 (1.36)
Rosal ³⁰	2011	US	RCT	124	128	5 to 10 years	5 to 10 years	>18 years	>18 years	Family members in the same home as patients	>10	>2	Both	The intervention consisted of 12 weekly and 8 monthly sessions. Cultural tailoring included the use of an educational soap opera to introduce self-management. The first session was conducted as an individual 1-h meeting in the participant's home. Group sessions lasted for approximately 2.5h. The intervention delivered by a trained team (either a nutritionist or health educator and trained lay individuals).	Usual care	12 month	8.39(1.29)	8.91 (0,80)
Toobert ³¹	2011	US	RCT	142	138	8.4	10.4	55.6	58.7	Not- mentioned	>10	>2	Group	The intervention delivered by physician consist of weekly meetings the intervention continued with 4-h facilitator-led meetings, providing 1 h each of instruction and practice of diet, stress management, physical activity, and support groups (1 h for each component-4 h). Meetings were held weekly for 6 months, then faded to twice monthly for 6 months	Usual care-only	12 month	8.3 (1.9)	8.3 (1.6)
Castejon ³²	2014	US	RCT	19	24			54	55	Family member or a friend Attending partners were a spouse, friend, sibling, or child	>10	1-2	Both	The intervention included a focused discussion (3- h) and two individual pharmacist counseling sessions (1-h) on medication, nutrition, exercise, and self-care to promote behavior changes. Sessions were culturally adapted for language, diet, family participation, and cultural beliefs. patients were asked to come back for three educational sessions every two weeks during the first six weeks and a follow- up clinical screening three months later. The family member was asked to come with the study participant for every clinical screening and educational session, as appropriate.	The control group had The same timeline but no education al sessions were given	3 month	7.3 (0.3)	88.0 (0.2)

	17	C 1	р ·	Samp	ole (N)	Dial durati	oetes on (yrs)	Mea	n age	Definition	Cor hour	ntact /frequ	Type of			Durati	Mean I	HbA1c
Author	Year	Country	Design	IG	CG	IG	CG	IG	CG	of Family	ei cor	ncy ntact	deliver y	Family involvement in DSME intervention	Control	on	IG (Mean/SD)	CG (Mean/SD)
Tabasi ³⁹	2014	Iran	RCT	45	46	9.71	11.39	52.93	54.13	Family members have blood relative and living with patient	<10	1	Group	The instruction about the importance of medication adherence and family support behavior was carried out about 45-60 minutes in 3 sessions. In every session 30-45 minutes considered for teaching and 15 minutes answering the questions and exchanging of views between family members.	Family did not receive any instructio n	3 month	7.7 (1.1)	8.1 (8.0)
Garcı´a ³³	2015	US	RCT	39	33	6.2	7.2	50	49.1	Not- mentioned	>10	>2	Individ ual	Participants received eight weekly, in-home, one- on-one educational and behavior modification sessions with a registered nurse focusing on symptom awareness, glucose self-testing and appropriate treatments, followed by eight biweekly support telephone sessions. each session lasting 30-60 min. Family members were encouraged to attend the sessions	Wait- listed control	6 month	7.9 (1.87)	8.5 (1.72)
Shakibaz adeh ³⁸	2015	Iran	RCT	140	140	5–9 years	5–9 years	58.73	58.57	First degree relative	>10	1-2	Group	A nurse educator, dietitian and counselor conducted the program. Eight 2½-h educational workshops offered over a 4-week period and followed by two sessions, each 2 weeks apart. Researcher set up group-based classes and invited a family member to enhance the self-efficacy of the patients and improve family support.	Usual Lecture- based non- interactiv e Educatio nal sessions.	18-21 month	8.1 (1.6)	8.9 (2.2)
Trief ³⁴	2016	US	RCT	104	82	12.8	12.6	57.8	56.9	Couples was defined as his/her partner in a committed relationship ≥ 1 year	>10	>2	Individ ual	A comprehensive Diabetes Education (DE) delivered in two telephone sessions (mean length of calls 75 min) followed with couple calls (cc). CC had 10 additional calls (mean length: CC 57 min/call). Couples were encouraged to provide mutual support for change. Calls occurred weekly for 12 weeks. Educators were dietitians	Usual care	12 month	8.5 (1.5)	8.5 (1.4)

	•	<u> </u>	р.;	Samj	ple (N)	Dia durati	betes on (yrs)	Mea	n age	Definition	Co hou	ontact r/frequ	Type of			Durati	Mean	HbA1c
Author	Year	Country	Design	IG	CG	IG	CG	IG	CG	of Family	e co	ency ntact	deliver y	Family involvement in DSME intervention	Control	on	IG (Mean/SD)	CG (Mean/SD)
Ing ³⁵	2016	Hawai	RCT	25	22	-	-	54.62	54.42	Not- mentioned	>10	>2	Group	1-hour weekly group meetings, providing information on diabetes self-management and encouraging participants to work with their diabetes team that includes the individual, their family, physician, and other diabetes experts	Usual care	6 month	8.96 (1.82)	9.47 (2.69)
Gomes ⁴⁴	2017	Brazil	RCT	108	114	15.7	7.82	60.43	60.43	A family member when patients do have blood relatives.	<10	1-2	Group	Participants received in four sessions with an average duration of 2 h each in groups of up to 10 people. a family caregiver who was considered a source of SS for the patient was included in the Education Program in Diabetes Mellitus	Usual care without family/ca regiver	12 month	8.73 (1.72)	8.94 (1.68)
McEwen ³⁶	2017	US	RCT	83	74	11.92	11.05	53.64	53.41	Lived in same house as participant with T2DM or saw them weekly	>10	>2	Both	The 12-week intervention program included 3 successive components: (1) six 2-hour educational and social support group sessions conducted weekly for 6 weeks, (2) three 2-hour home visits scheduled weekly for 3 weeks, and (3) three 20-minute telephone calls scheduled weekly for 3 weeks. A nurse who is a certified diabetes educator (CDE) conducted the educational sessions, and a promotor conducted the social support sessions, home visits, and telephone calls.	Waitlist control group	9 month	9.19 (2.1)	9,20 (2.0)
Wichit ⁴⁰	2017	Thailand	rct	70	70	6.0	5.4	61.3	55.5	Living in the same residence and relative,	<10	1	Group	The intervention group received routine care plus a family-oriented program that included education classes, group discussions, a home visit, and a telephone follow-up. The education consisted 2 hour sessions were provided in a group of 8-12 dyads (individual and family member) per group and the facilitator. The program consisted of three education sessions delivered at baseline, Week 5, and Week 9. The facilitator of the education session was a Thai National and a registered nurse.	Usual care rroutine care	13 Week	7.0 (1.2)	7.3 (1.4)

A 4h	V	Compten	Destau	Sam	ple (N)	Dia durat	ibetes ion (yrs)	Mea	n age	Definition	Co hour	ntact ∵/frequ	Type of	Family inclusion of in DOME in America	Control	Durati	Mean	HbA1c
Autnor	y ear	Country	Design	IG	CG	IG	CG	IG	CG	of Family	e coi	ncy ntact	deliver y	Family involvement in DSME intervention	Control	on	IG (Mean/SD)	CG (Mean/SD)
Withidpan yawong ⁴¹	2018	Thailand	RCT	98	98	5.61	6.35	60.53	58.13	Living in the same Household as or significant relative of the participant	<10	1	Individ ual	Beyond the usual care, the intervention group received an education package for participants and their relatives. The intervention was administered by one research pharmacist during 4 visits within a 9-month period, at approximately 3-month intervals. Each intervention lasted 40-50. A pharmacist delivered the educational sessions and encouraged family members to take an active role in self-management practices.	Usual care	9 month	7.84 (1.96)	8.87 (1.81)
McElfish 37	2019	US	RCT	110	111	-	-	52.2	52.2	Same households	>10	1-2	Individ ual	The adapted DSME included 10 h of content delivered over an 8-week period and covered eight core elements of DSME. the adapted DSME curriculum engaged family members in the educational sessions delivered by certified educators.	Usual care without family members	12 month	9.64 (1.74)	10.36 (1.85)

Meta-analysis results

A meta-analysis was conducted using data from the final sample size of 17 studies with 2644 participants. Compared to the control group, Figure 2 displays the HbA1c forest plot of the DSME intervention with family involvement. The effect size obtained was -0.31 (95% CI -0.46 to -0.16; 2644 participants; 17 study studies). The heterogeneity was high ($I^2 =$ 71%), and it was statistically significant (p<.001). The random-effects metaanalysis was used to evaluate differences between the intervention and control groups in the baseline HbA1c. Compared to the control group, there was also a significant difference in HbA1c in the intervention group. There was no evidence for publication bias in the funnel plot (Figure. 3), the Egger's test (p = .123), or the Begg's test (p = .458).

	Family Invo	olvement D	SME	Usu	al Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
1.1.1 Overall									
Brown 2002	10.89	2.56	126	11.64	2.85	126	1.2%	-0.28 [-0.52, -0.03]	
Castejon 2014	7.3	0.3	19	8	0.2	24	0.4%	-2.76 [-3.62, -1.90]	←
Garcia 2015	7.9	1.87	39	8.5	1.72	33	0.8%	-0.33 [-0.80, 0.14]	
Gomes 2017	8.73	1.72	108	8.94	1.68	114	1.2%	-0.12 [-0.39, 0.14]	-+
Ing 2016	8.96	1.82	25	9.47	2.69	22	0.6%	-0.22 [-0.80, 0.35]	
Kang 2010	7.9	1.36	33	8.12	1.21	34	0.8%	-0.17 [-0.65, 0.31]	
Keogh 2011	8.41	0.99	60	8.8	1.36	61	1.0%	-0.33 [-0.68, 0.03]	
McElfish 2019	9.64	1.74	110	10.36	1.85	111	1.2%	-0.40 [-0.67, -0.13]	
McEwen 2017	9.19	2.1	83	9.2	2	74	1.1%	-0.00 [-0.32, 0.31]	
Rosal 2011	8.39	1.29	124	8.91	0.8	128	1.2%	-0.48 [-0.74, -0.23]	
Shakibazadeh 2015	8.1	1.6	140	8.9	2.2	140	1.3%	-0.41 [-0.65, -0.18]	
Tabasi 2014	7.7	1.1	45	8.1	8	46	0.9%	-0.07 [-0.48, 0.34]	
Toobert 2011	8.3	1.9	142	8.3	1.6	138	1.3%	0.00 [-0.23, 0.23]	
Trief 2016	8.5	1.5	104	8.5	1.4	82	1.1%	0.00 [-0.29, 0.29]	
Vincent 2007	6.14	0.5	9	6.84	1.3	8	0.3%	-0.69 [-1.68, 0.30]	
Wichit 2016	7	1.2	70	7.3	1.4	70	1.1%	-0.23 [-0.56, 0.10]	
Withidpanyawong 2018	7.84	1.96	98	8.87	1.81	98	1.2%	-0.54 [-0.83, -0.26]	
Subtotal (95% CI)			1335			1309	16.7%	-0.31 [-0.46, -0.16]	\bullet
Heterogeneity: Tau ² = 0.07	; Chi² = 55.4	5, df = 16 (P < 0.00	001); P	= 71%				
Test for overall effect: Z = 4	1.00 (P < 0.00	001)							-Z -1 U 1 Z
									Favours (Family Dowe) Favours (Usual Care)

Figure 2 Forest plot for HbA1c meta-analysis



Figure 3 Funnel plot

Subgroup analysis

Subgroup analysis was conducted to explore possible differences between studies based on their characteristics, participants' baseline HbA1c levels, intervention duration, type of intervention delivery, contact hour, and contact frequency (Table. 2).

Particularly, a subgroup analysis of HbA1c values to explore the source of

heterogeneity between studies was carried out to provide a more detailed overview of the findings. Subgroup analysis of HbA1c values was performed based on the baseline HbA1c grouping (<9% with \geq 9%), intervention duration (<6 months with \geq 6 months), contact hour (<10 with \geq 10), contact frequency (low, moderate, high), and program delivery (individual, group, and combination).

		Std Mean			
Variables	No Studies	Difference of HbA1c	95% CI	р	I ²
All studies	17	-0.31	-0.46 to -0.16	<.001	71%
Baseline HbA1c					
< 9%	6	-0.55	-1.05 to -0.05	.030	87%
$\geq 9\%$	11	-0.29	-0.40 to -0.17	<.001	37%
Duration of					
intervention					
< 6 month	4	-0.87	-1.78 to 0.04	.060	91%
\geq 6 month	13	-0.26	-0.37 to -0.15	<.001	44%
Type of intervention					
delivery	5	0.22	0.52 ± 0.12	001	16.
Individual	3	-0.32	-0.52 to -0.12	.001	40%
Group	8	-0.20	-0.32 to -0.09	<.001	13%
Combination	4	-0.73	-1.42 to -0.03	.040	92%
Contact Hour					
< 10	6	-0.26	-0.41 to -0.11	<.001	16%
≥10	11	-0.37	-0.59 to -0.14	.001	80%
Frequency of Contact					
Low	4	-0.29	-0.51 to -0.07	.010	33%
Moderate	4	-0.75	-1.39 to -0.12	.020	91%
High	9	-0.23	-0.38 to -0.08	.002	49%

Table 2 Subgroup analysis for the standard mean difference of HbA1c

Subgroup analysis based on the baseline HbA1c

A subgroup analysis was performed for the baseline HbA1c <9% and \geq 9%. statistically significant There was difference in baseline HbA1c <9% (SMD= -0.55; 95%CI -1.05 to -0.05; p=.030) and \geq 9% (SMD= -0.29; 95%CI 0.40 to -0.17; p<.001). Study heterogeneity was high and statistically significant for the baseline HbA1c <9% (I² = 87%; p< .001), but for the baseline HbA1c $\geq 9\%$, it was low and statistically non-significant ($I^2 = 37\%$; p= .10) (Figure. 4). There was evidence of potential publication bias according to Egger's test (p= .078) and Begg's test (p=.038) for the baseline HbA1c <9%. Meanwhile, there was no evidence of publication bias for the baseline HbA1c \geq 9% according to Egger's test (p=.409) and Begg's test (p=.311). Previous studies indicated that less than 9 percent of baseline HbA1c was associated with greater improvement in glycemic control interventions than more than 9 percent of baseline HbA1c⁴⁶.

Subgroup analysis based on the intervention duration

A subgroup analysis was then conducted for the <6 months and ≥6 -month of intervention duration. There was no statistically significant difference in for HbA1c <6-month duration of intervention (SMD=-0.87: 95%CI -1.78 to 0.04; p=.060), but there was statistically significant difference in HbA1c for ≥ 6 month duration of intervention (SMD= -0.26; 95%CI -0.37 to -0.15; p<.001). Study heterogeneity was high and statistically significant for the <6-month duration of the intervention ($I^2 = 91\%$; p<.001), but low and statistically significant for the ≥ 6 month duration of the intervention $(I^2 =$ 44%; p=004) (Figure. 5). There was no evidence of publication bias for <6-month duration of intervention according to Egger's test (p= .296) and Begg's test

(p=.496); however, there was no evidence of publication bias for \geq the 6-month duration of intervention according to Egger's test (p= .846) and Begg's test (p=.807). The intervention duration was related to glycemic control. Previous studies have found that interventions carried out for six months resulted in significant changes in blood sugar control⁴⁷.

Subgroup analysis based on the type of intervention delivery

A subgroup analysis was carried out based on the type of intervention delivery for individual, group, and combination. There were statistically significant differences in HbA1c for individual delivery (SMD= -0.32; 95%CI -0.52 to -0.12; p=.001), group delivery (SMD= -0.20; 95%CI -0.32 to -0.09; p<.001), and combination delivery (SMD=-0.73; 95%CI -1.42 to -0.03; p=.040). Study heterogeneity was low and statistically non-significant for the individual delivery $(I^2 = 46\%; p=.12)$ and group delivery $(I^2 =$ 13%; p=.33), but it was high and statistically significant for the combination delivery ($I^2 = 92\%$; p<.001) (Figure. 6). There was no evidence of publication biases for individual delivery according to Egger's test (p= .933) and Begg's test (p=.624), for group delivery based on Egger's test (p= .644) and Begg's test (p=.804), and for combination delivery as shown by Egger's test (p=.356) and Begg's test (p=.496).

Subgroup analysis based on the contact hours

A subgroup analysis was done for contact hours <10 and \geq 10. There were statistically significant differences in HbA1c for contact hours <10 (SMD= -0.26; 95%CI -0.41 to -0.11; p<.001) and contact hours \geq 10 (SMD= -0.37; 95%CI -0.59 to -0.14; p= .001). Study heterogeneity was low and statistically non-significant for contact hours <10 ($I^2 = 16\%$; p=.031), but for contact hours ≥ 10 , it was high and statistically significant ($I^2 = 80\%$; p<.001) (Figure. 7). There were no evidence of publication biases for contact hours <10 according to Egger's test (p=.604) and Begg's test (p=.851) and for contact hours ≥ 10 based on Egger's test (p=.126) and Begg's test (p=.483).

Subgroup analysis based on the contact frequency

A subgroup analysis was conducted based on low, moderate, and high contact frequency. There were statistically significant differences in HbA1c for low contact frequency (SMD= -0.29; 95%CI -0.51 to -0.07; p=.010), moderate contact frequency (SMD= -0.75; 95%CI -1.39 to - 0.12; p=.020), and high contact frequency (SMD= -0.23; 95%CI -0.38 to -0.08; p=.002). Study heterogeneity was low and statistically non-significant for the low contact frequency ($I^2 = 33\%$; p=.12), high statistically significant for the and moderate contact frequency ($I^2 = 91\%$; low and p<.001), and statistically significant for the high contact frequency $(I^2 = 49\%; p=.002)$ (Figure. 8). There was no evidence of publication biases for the low contact frequency according to Egger's test (p=.183) and Begg's test (p=.174), for moderate contact frequency based on Egger's test (p= .073) and Begg's test (p=.496), and for a high contact frequency as found in Egger's test (p= .501) and Begg's test (p=.297).

	Family Inv	olvement	DSME	Usu	al Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
1.1.2 Baseline HbA1c < 9									
Castejon 2014	7.3	0.3	19	8	0.2	24	0.4%	-2.76 [-3.62, -1.90] 🔸	·
Garcia 2015	7.9	1.87	39	8.5	1.72	33	0.8%	-0.33 [-0.80, 0.14]	
Tabasi 2014	7.7	1.1	45	8.1	8	46	0.9%	-0.07 [-0.48, 0.34]	
Toobert 2011	8.3	1.9	142	8.3	1.6	138	1.3%	0.00 [-0.23, 0.23]	
Vincent 2007	6.14	0.5	9	6.84	1.3	8	0.3%	-0.69 [-1.68, 0.30]	
Wichit 2016	7	1.2	70	7.3	1.4	70	1.1%	-0.23 [-0.56, 0.10]	
Subtotal (95% CI)			324			319	4.7%	-0.55 [-1.05, -0.05]	\bullet
Heterogeneity: Tau ² = 0.31	; Chi ² = 38.6	i2, df = 5 (F	° < 0.000	01); I² =	87%				
Test for overall effect: Z = 2	.17 (P = 0.0	3)							
1.1.3 Baseline HbA1c >=9									
Brown 2002	10.89	2.56	126	11.64	2.85	126	1.2%	-0.28 [-0.52, -0.03]	
Gomes 2017	8.73	1.72	108	8.94	1.68	114	1.2%	-0.12 [-0.39, 0.14]	
Ing 2016	8.96	1.82	25	9.47	2.69	22	0.6%	-0.22 [-0.80, 0.35]	
Kang 2010	7.9	1.36	33	8.12	1.21	34	0.8%	-0.17 [-0.65, 0.31]	
Keogh 2011	8.41	0.99	60	8.8	1.36	61	1.0%	-0.33 [-0.68, 0.03]	
McElfish 2019	9.64	1.74	110	10.36	1.85	111	1.2%	-0.40 [-0.67, -0.13]	
McEwen 2017	9.19	2.1	83	9.2	2	74	1.1%	-0.00 [-0.32, 0.31]	
Rosal 2011	8.39	1.29	124	8.91	0.8	128	1.2%	-0.48 [-0.74, -0.23]	
Shakibazadeh 2015	8.1	1.6	140	8.9	2.2	140	1.3%	-0.41 [-0.65, -0.18]	
Trief 2016	8.5	1.5	104	8.5	1.4	82	1.1%	0.00 [-0.29, 0.29]	
Withidpanyawong 2018	7.84	1.96	98	8.87	1.81	98	1.2%	-0.54 [-0.83, -0.26]	
Subtotal (95% CI)			1011			990	12.0%	-0.29 [-0.40, -0.17]	◆
Heterogeneity: Tau ² = 0.01	; Chi ² = 16.0	10, df = 10	(P = 0.10)); I ² = 37	%			_	
Test for overall effect: Z = 4	.89 (P < 0.0	0001)							Favours [Family DSME] Favours [Usual Care]

Figure 4 Forest plot for Subgroup analysis based on the HbA1c baseline

Vol.19 No.3 September-December 2021

	Consilu Inus	humant D.C.		llau				Stal Maan Difference	Ctd Mann Difference
Chudu an Cubanaun	Family Invo	ivement DSI		USU	arcar	Tatal	Mainhé	Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	lotal	Mean	SD	lotal	weight	IV, Random, 95% CI	IV, Random, 95% CI
1.1.4 Duration of Intervent	ion < 6 mont	th							
Castejon 2014	7.3	0.3	19	8	0.2	24	0.4%	-2.76 [-3.62, -1.90]	←
Tabasi 2014	7.7	1.1	45	8.1	8	46	0.9%	-0.07 [-0.48, 0.34]	
Vincent 2007	6.14	0.5	9	6.84	1.3	8	0.3%	-0.69 [-1.68, 0.30]	
Wichit 2016	7	1.2	70	7.3	1.4	70	1.1%	-0.23 [-0.56, 0.10]	
Subtotal (95% CI)			143			148	2.6%	-0.87 [-1.78, 0.04]	
Heterogeneity: Tau ² = 0.75	; Chi ² = 32.89	9, df=3 (P ≺	0.000	01); I ² =	91%				
Test for overall effect: Z = 1	.87 (P = 0.06)							
1.1.5 Duration of Intervent	ion >=6 mon	th							
Brown 2002	10.89	2.56	126	11.64	2.85	126	1.2%	-0.28 [-0.52, -0.03]	
Garcia 2015	7.9	1.87	39	8.5	1.72	33	0.8%	-0.33 [-0.80, 0.14]	
Gomes 2017	8.73	1.72	108	8.94	1.68	114	1.2%	-0.12 [-0.39, 0.14]	
Ing 2016	8.96	1.82	25	9.47	2.69	22	0.6%	-0.22 [-0.80, 0.35]	
Kang 2010	7.9	1.36	33	8.12	1.21	34	0.8%	-0.17 [-0.65, 0.31]	
Keogh 2011	8.41	0.99	60	8.8	1.36	61	1.0%	-0.33 [-0.68, 0.03]	
McElfish 2019	9.64	1.74	110	10.36	1.85	111	1.2%	-0.40 [-0.67, -0.13]	
McEwen 2017	9.19	2.1	83	9.2	2	74	1.1%	-0.00 [-0.32, 0.31]	
Rosal 2011	8.39	1.29	124	8.91	0.8	128	1.2%	-0.48 [-0.74, -0.23]	
Shakibazadeh 2015	8.1	1.6	140	8.9	2.2	140	1.3%	-0.41 [-0.65, -0.18]	
Toobert 2011	8.3	1.9	142	8.3	1.6	138	1.3%	0.00 [-0.23, 0.23]	
Trief 2016	8.5	1.5	104	8.5	1.4	82	1.1%	0.00 [-0.29, 0.29]	
Withidpanyawong 2018	7.84	1.96	98	8.87	1.81	98	1.2%	-0.54 [-0.83, -0.26]	
Subtotal (95% CI)			1192			1161	14.0%	-0.26 [-0.37, -0.15]	•
Heterogeneity: Tau ² = 0.02	; Chi ² = 21.39	9, df = 12 (P =	= 0.04)); I ^z = 44	%				
Test for overall effect: Z = 4	.50 (P < 0.00	001)							Favours [Family DSME] Favours [Usual Care]

Figure 5 Forest plot for Subgroup analysis based on duration of intervention

	Family Invo	olvement [OSME	Usu	ial Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
1.1.6 Type of delivery indi	vidual								
Garcia 2015	7.9	1.87	39	8.5	1.72	33	0.8%	-0.33 [-0.80, 0.14]	
Keogh 2011	8.41	0.99	60	8.8	1.36	61	1.0%	-0.33 [-0.68, 0.03]	
McElfish 2019	9.64	1.74	110	10.36	1.85	111	1.2%	-0.40 [-0.67, -0.13]	
Trief 2016	8.5	1.5	104	8.5	1.4	82	1.1%	0.00 [-0.29, 0.29]	
Withidpanyawong 2018	7.84	1.96	98	8.87	1.81	98	1.2%	-0.54 [-0.83, -0.26]	
Subtotal (95% CI)			411			385	5.3%	-0.32 [-0.52, -0.12]	◆
Heterogeneity: Tau ² = 0.00	2; Chi ² = 7.40	, df = 4 (P =	= 0.12); P	² = 46%					
Test for overall effect: Z = 3	3.21 (P = 0.0)	01)							
1.1.7 Type of deliverygrou	ıp								
Brown 2002	10.89	2.56	126	11.64	2.85	126	1.2%	-0.28 [-0.52, -0.03]	
Gomes 2017	8.73	1.72	108	8.94	1.68	114	1.2%	-0.12 [-0.39, 0.14]	-+
Ing 2016	8.96	1.82	25	9.47	2.69	22	0.6%	-0.22 [-0.80, 0.35]	
Shakibazadeh 2015	8.1	1.6	140	8.9	2.2	140	1.3%	-0.41 [-0.65, -0.18]	
Tabasi 2014	7.7	1.1	45	8.1	8	46	0.9%	-0.07 [-0.48, 0.34]	
Toobert 2011	8.3	1.9	142	8.3	1.6	138	1.3%	0.00 [-0.23, 0.23]	_
Vincent 2007	6.14	0.5	9	6.84	1.3	8	0.3%	-0.69 [-1.68, 0.30]	
Wichit 2016	7	1.2	70	7.3	1.4	70	1.1%	-0.23 [-0.56, 0.10]	
Subtotal (95% CI)			665			664	7.9%	-0.20 [-0.32, -0.09]	\blacklozenge
Heterogeneity: Tau ² = 0.00	0; Chi² = 8.02	, df = 7 (P =	= 0.33); P	² =13%					
Test for overall effect: Z = 3	3.38 (P = 0.0)	007)							
1.1.9 Type of delivery bot	h								
Castejon 2014	7.3	0.3	19	8	0.2	24	0.4%	-2.76 [-3.62, -1.90]	←
Kang 2010	7.9	1.36	33	8.12	1.21	34	0.8%	-0.17 [-0.65, 0.31]	
McEwen 2017	9.19	2.1	83	9.2	2	74	1.1%	-0.00 [-0.32, 0.31]	
Rosal 2011	8.39	1.29	124	8.91	0.8	128	1.2%	-0.48 [-0.74, -0.23]	
Subtotal (95% CI)			259			260	3.5%	-0.73 [-1.42, -0.03]	
Heterogeneity: Tau ² = 0.44	4; Chi² = 36.4	6, df = 3 (P	o.000 × ۹	01); I ² =	92%				
Test for overall effect: Z = :	2.06 (P = 0.04	4)							-2 -1 U 1 2 Eavoure [Eamily DSME] Eavoure [Usual Care]
									Favours (Family Dowe) Favours (Osual Care)

Figure 6 Forest plot for Subgroup analysis based on type of intervention delivery

Vol.19 No.3 September-December 2021

	Family Invo	olvement D	OSME	Usu	al Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
1.1.10 Contact hour <10									
Gomes 2017	8.73	1.72	108	8.94	1.68	114	1.2%	-0.12 [-0.39, 0.14]	-+
Kang 2010	7.9	1.36	33	8.12	1.21	34	0.8%	-0.17 [-0.65, 0.31]	
Keogh 2011	8.41	0.99	60	8.8	1.36	61	1.0%	-0.33 [-0.68, 0.03]	
Tabasi 2014	7.7	1.1	45	8.1	8	46	0.9%	-0.07 [-0.48, 0.34]	
Wichit 2016	7	1.2	70	7.3	1.4	70	1.1%	-0.23 [-0.56, 0.10]	
Withidpanyawong 2018	7.84	1.96	98	8.87	1.81	98	1.2%	-0.54 [-0.83, -0.26]	<u> </u>
Subtotal (95% CI)			414			423	6.1%	-0.26 [-0.41, -0.11]	◆
Heterogeneity: Tau ² = 0.01	; Chi² = 5.97	, df = 5 (P :	= 0.31); P	² = 16%					
Test for overall effect: Z = 3	.40 (P = 0.00	307)							
1.1.11 Contact hour .>= 10)								
Brown 2002	10.89	2.56	126	11.64	2.85	126	1.2%	-0.28 [-0.52, -0.03]	
Castejon 2014	7.3	0.3	19	8	0.2	24	0.4%	-2.76 [-3.62, -1.90] 🔸	
Garcia 2015	8.1	1.6	140	8.9	2.2	140	1.3%	-0.41 [-0.65, -0.18]	
Ing 2016	8.96	1.82	25	9.47	2.69	22	0.6%	-0.22 [-0.80, 0.35]	
McElfish 2019	9.64	1.74	110	10.36	1.85	111	1.2%	-0.40 [-0.67, -0.13]	
McEwen 2017	9.19	2.1	83	9.2	2	74	1.1%	-0.00 [-0.32, 0.31]	
Rosal 2011	8.39	1.29	124	8.91	0.8	128	1.2%	-0.48 [-0.74, -0.23]	
Shakibazadeh 2015	7.9	1.87	39	8.5	1.72	33	0.8%	-0.33 [-0.80, 0.14]	
Toobert 2011	8.3	1.9	142	8.3	1.6	138	1.3%	0.00 [-0.23, 0.23]	
Trief 2016	8.5	1.5	104	8.5	1.4	82	1.1%	0.00 [-0.29, 0.29]	
Vincent 2007	6.14	0.5	9	6.84	1.3	8	0.3%	-0.69 [-1.68, 0.30]	
Subtotal (95% CI)			921			886	10.6%	-0.37 [-0.59, -0.14]	\bullet
Heterogeneity: Tau ² = 0.10	; Chi ² = 49.4	5, df = 10 ((P < 0.00	001); P	= 80%			-	
Test for overall effect: Z = 3	.20 (P = 0.00	DI)							-2 -1 U 1 2 Foreuro (Formily DOME) - Foreuro (Lloud) Corol
									Favours (Family DSME) Favours (Osual Care)

Figure 7 Forest plot for Subgroup analysis based on contact hour

	Family Invo	lvement C	SME	Usu	al Car	е		Std. Mean Difference	Std. Mean Difference					
Study or Subgroup	Mean	\$D	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl					
1.1.12 Frequency of conta	actless													
Kang 2010	7.9	1.36	33	8.12	1.21	34	0.8%	-0.17 [-0.65, 0.31]						
Tabasi 2014	7.7	1.1	45	8.1	8	46	0.9%	-0.07 [-0.48, 0.34]						
Wichit 2016	7	1.2	70	7.3	1.4	70	1.1%	-0.23 [-0.56, 0.10]	+					
Withidpanyawong 2018	7.84	1.96	98	8.87	1.81	98	1.2%	-0.54 [-0.83, -0.26]						
Subtotal (95% CI)			246			248	3.9%	-0.29 [-0.51, -0.07]	◆					
Heterogeneity: Tau ² = 0.02	; Chi ² = 4.46,	df = 3 (P =	= 0.22); P	² = 33%										
Test for overall effect: Z = 2.57 (P = 0.01)														
1.1.13 Frequency of contact moderate														
Casteinn 2014	7.3	0.3	19	8	0.2	24	0.4%	-2 76 [-3 62 -1 90]	←					
Gomes 2017	8.73	1.72	108	8.94	1.68	114	1.2%	-0.12 [-0.39, 0.14]						
McElfish 2019	9.64	1.74	110	10.36	1.85	111	1.2%	-0.40 [-0.67, -0.13]						
Shakibazadeh 2015	7.9	1.87	39	8.5	1.72	33	0.8%	-0.33 [-0.80, 0.14]	<u> </u>					
Subtotal (95% CI)			276			282	3.6%	-0.75 [-1.39, -0.12]						
Heterogeneity: Tau ² = 0.36	; Chi ² = 33.2-	4, df = 3 (P	< 0.000	01); I ^z =	91%									
Test for overall effect: Z = 2	2.33 (P = 0.02)												
1.1.14 Frequency of conta	act high													
Brown 2002	10.89	2.56	126	11.64	2.85	126	1.2%	-0.281-0.52-0.031						
Garcia 2015	8.1	1.6	140	8.9	2.2	140	1.3%	-0.41 [-0.65, -0.18]						
Ing 2016	8.96	1.82	25	9.47	2.69	22	0.6%	-0.22 [-0.80, 0.35]						
Keogh 2011	8.41	0.99	60	8.8	1.36	61	1.0%	-0.33 [-0.68, 0.03]						
McEwen 2017	9.19	2.1	83	9.2	2	74	1.1%	-0.00 [-0.32, 0.31]						
Rosal 2011	8.39	1.29	124	8.91	0.8	128	1.2%	-0.48 [-0.74, -0.23]						
Toobert 2011	8.3	1.9	142	8.3	1.6	138	1.3%	0.00 [-0.23, 0.23]	+					
Trief 2016	8.5	1.5	104	8.5	1.4	82	1.1%	0.00 [-0.29, 0.29]						
Vincent 2007	6.14	0.5	9	6.84	1.3	8	0.3%	-0.69 [-1.68, 0.30]						
Subtotal (95% CI)			813			779	9.2%	-0.23 [-0.38, -0.08]	\bullet					
Heterogeneity: Tau ² = 0.02	; Chi² = 15.6	6, df = 8 (P	= 0.05);	l² = 499	6									
Test for overall effect: Z = 3	3.07 (P = 0.00	12)							Favours (Family DSME) Favours (Usual Care)					

Figure 8 Forest plot for Subgroup analysis based on frequency of contact

DISCUSSION

This study's primary purpose was to review studies on improving glycemic control using family involvement in diabetes self-management education (DSME). The researchers conducted a meta-analysis to estimate intervention effects and identify variables contributing to glycemic control in type 2 DM. In this meta-analysis, only randomized control trials were included. This review included seventeen randomized control trial studies with 2644 participants. This review's results indicated that family involvement in DSME intervention positively affected HbA1c levels in type 2 diabetes.

A total of seven articles showed a significant value in reducing HbA1c to control^{28,30,32,37,38,41,42}. This meta-analysis suggested that family involvement in DSNE had a favorable effect on improving glycemic control, with a pooled mean reduction of -0,31 (95% CI -0.46 to -0.16) in HbA1c levels compared with usual care, and it was statistically significant (p < .001). This finding is consistent with prior systematic review and meta-analyses about group-based DSME compared to routine treatment, which found HbA1c reductions in the general population⁴⁸. The estimated -0.31% reduction in HbA1c in patients with type 2 diabetes was modest. There is also some evidence from observational studies, which disclosed a decrease in HbA1c reduced the risk of complications. According to a prior study, each 1% reduction in HbA1c was associated with reductions in risk of 21% for any endpoint related to diabetes, 21% for deaths related to diabetes, 14% for myocardial infarction, and 37% for microvascular complications⁴⁹.

This study has essential implications for current clinical and public health practice and research. Glycemic control is a vital predictor of many chronic complications of diabetes. Another study stated that every 1% reduction in HbA1c was associated with 14%, 12%, and 16% reductions in the relative risk (RR) of myocardial infarction, stroke, and heart respectively⁵⁰. failure, Moreover, according to the prior study, significantly increased risks of complications for every 1% higher HbA1c level were associated with a 38% higher risk of a macrovascular event, a 40% higher risk of a microvascular event, and a 38% higher risk of death⁵¹.

In this study, there was a high value of heterogeneity, and it was statistically significant (I²= 71%; P<.001). This high heterogeneity value might be due to the substantial variations in intervention characteristics. The random-effects model was used because of this heterogeneity value. Various subgroup studies were conducted not only to clarify the heterogeneity (I²) between studies and identify the most effective intervention aspects but also to explore the effects. The current study indicated that larger effects were found in interventions with a longer duration and contact hours, delivered in the combination between individual and group format, and with interventions of lower baseline HbA1c and interventions of a moderate contact frequency.

Subgroup analysis revealed that studies involving patients with lower baseline HbA1c (< 9%) showed a higher greater effect size (SMD= -0.55) after intervention than \geq 9% of baseline HbA1c. The results are contrary to the previous study, which found that a higher baseline HbA1c (< 9%) was reported, a greater decrease in HbA1c^{52,53}. The low baseline HbA1c in the other study had the best clinical predictor of HbA1c attainment at 7.0%. A higher risk of hypoglycemia was also associated with the low baseline HbA1c⁵⁴. Efforts to reach the HbA1c target should be balanced with quality of life preservation and protection against excessive hypoglycemia, according to the ADA statement⁵⁵.

Besides, the results from subgroup analysis based on the intervention duration showed that the ≥ 6 months effectively reduced HbA1c, and it was statistically significant (p<.001). There was evidence to suggest that a longer intervention duration was associated with a reduction in HbA1c levels. In a previous review, outcome measures were primarily collected at 6 and 12 months. The HbA1c reduction results varied. Of the seven studies, a decrease in HbA1c levels was found in four studies, while the other three studies did not show a significant decrease in HbA1c levels⁵⁶. The factors responsible could hardly be identified because they limited the available knowledge about the interventions' long-term effects and needed additional research.

Further, the chosen studies were divided into three subgroups based on the type of intervention delivery, including individual, group, and a combination of individual and group. In this current review, the effect size between the combination between individual and group delivery was greater than that between the individual or group delivery only. The entire category was statistically significant. The current findings are consistent with the previous review, which uncovered that a combination of group and individual involvement resulted in the largest decreases in A1C than only group and individual⁵⁷. A combination of individual education can provide and group interactive, participatory, and collaborative learning to positively affect health outcomes from planning and problem solving and improving health behavior.

Based on intervention contact hours between < 10 hours and 10 hours, subgroup analyses were also conducted. In this study, contact hour >10 had a greater effect size than < 10 hours in reducing HbA1c, and it statistically significant (p=.001). was Similar to these findings, the previous meta-analysis study suggested that the reduction in HbA1c values was associated with more contact hours, and it was statistically significant 57. The previous review also noted that contact hours of >10 hours had a major effect on reducing the risk of mortality and improving health outcomes^{58,59}.

Moreover, the subgroup analysis based on the contact frequency was classified into three levels: low (less than one contact per month per patient), moderate (one or two contacts per month per patient), and high (more than two contacts in a month per patient). The highest effect size in HbA1c was observed in the moderate category. All contact frequency categories were statistically significant. These results corroborate with the previous meta-analysis study, stating that moderate and high contact frequency intervention programs indicated а significant glycemic control improvement. Contact frequency seemed to be a vital feature of intervention efficacy. Therefore, a program with a moderate or intensive contact frequency should be implemented⁶⁰.

As explained, this meta-analysis used GRADE to assess quality evidence of the effect of the family involvement in DSME on glycemic control in patients with type 2 diabetes. According to GRADE, there was moderate-quality evidence for the effects of family involvement in DSME on glycemic control in patients with type 2 diabetes, and the reasons for providing the moderate quality of evidence among the four levels (high, moderate, low, and very low) were the possible risk of performance bias. The risk from performance bias was unavoidable as it is very difficult to provide allocation concealment and blind the patients and provider to random studies assignment, although few successfully blinded participants to randomization.

This methodological quality assessment result is in line with the previous meta-analysis. Winkley et al. found that the quality of evidence in studies involving adults with type 2 diabetes was rated as moderate quality for the primary outcome (HbA1c). The reasons for moderate quality were the inconsistency and high heterogeneity in most subgroups of psychological interventions and interventionist subgroups. Otherwise, the quality of evidence was rated as high quality for the secondary outcomes, such as BMI and blood pressure⁶¹.

Strengths and limitations

The study's strengths are а comprehensive systematic review and meta-analysis to examine the effect of family involvement in DSME on glycemic control. Only randomized controlled trials were included in this analysis to assure high validity. The researchers selected studies by two independent reviewers. This study is the first empirical review to explore the effectiveness of family involvement in DSME on glycemic control to the best of researchers' knowledge. the The researchers also performed subgroup analysis to answer clinically relevant and essential questions that had not previously been addressed. For all risks of bias assessed, most studies had a low risk of bias, and the overall body of evidence was rated to be moderate quality per the GRADE criteria.

This meta-analysis has several limitations. First, the researchers limited the selection of publications only to articles in the English and Indonesian language. Second, only peer-reviewed and published papers were evaluated. There was no assessment of grey literature, unpublished work, or dissertation studies. Third, the number of studies is frequently limited by the selective publication such as language, so only English and Indonesian articles were included in the meta-analysis. Fourth, there were considerable variations found in the intervention programs' components.

Furthermore, if the components between the trials were similar, variations could exist considering the intervention's strength or that the providers' education techniques would vary. Such variations might have contributed to the observed heterogeneity, reflected by the large CIs and I^2 values. There was a lack of clear definitions of "usual care" in most of the studies included, and there were also differences in the DSME methods in various studies. Although there were differences in the diabetes treatment methods provided for patients, it is clear that patients assigned to family DSME groups received more diabetes education than those in control groups.

CONCLUSION

In conclusion, the current evidence suggests that family involvement in DSME improves glycemic control in patients with type 2 diabetes. There are indications that less than 9% of baseline HbA1c, interventions delivered by individual and group combination, delivered in more than six months, contact hours longer than 10 hours, and moderate to high contact frequency give the best glycemic control results.

RECOMMENDATION

Further research with baseline HbA1c, intervention duration, delivery type, contact hour, contact frequency, and other characteristics can shed more light on the subject and provide useful information about DSME that will help determine the most successful regime. Secondly, research heterogeneity was high in this metaanalysis, indicating the need for more rigorously designed family DSME trials for patients with type 2 diabetes in the future. Thirdly, as in all meta-analyses, the risk of publication and selection bias must be considered. Finally, since the number of studies on family-based DSME is limited, further research with a large number of participants and a longer time of follow-up is required to gather further evidence.

ACKNOWLEDGMENT

The authors wish to thank Prof. Ari Natalia Probandari, dr., MPH, Ph.D. from Universitas Sebelas Maret, Surakarta, Indonesia, for her helpful advice on this study.

AUTHOR CONTRIBUTION

AA, BM, and DT had conceived the meta-analysis idea and were guarantors of the general content. AA and BM selected appropriate papers, retrieved them, and evaluated each trial. AA was responsible for the statistical analysis. AA and DT drafted the manuscript. All the authors revised the final manuscript and approved it.

CONFLICT OF INTEREST

There was no conflict of interest in this study.

THE AUTHORS DECLARE THAT THEY HAVE NO COMPETING INTERESTS.

None of the funding organizations or sponsors had any role in this study. This study was funded independently by authors.

ETHICAL APPROVAL

This article does not contain any studies with human participants performed by any of the authors.

REFFERENCE

- Onyango EM, Onyango BM. The rise of noncommunicable diseases in Kenya: An examination of the time trends and contribution of the changes in diet and physical inactivity. J Epidemiol Glob Health [Internet]. 2018 [cited 2021 January 29];8(1–2):1–7. Available from: https://pubmed.ncbi. nlm.nih.gov/30859780/
- 2. Peros JP. Diabetes self-management education (DSME) program for

glycemic control. University of San Diego; 2016.

- Khan MAB, Hashim MJ, King JK, Govender RD, Mustafa H, Kaabi J Al. Epidemiology of Type 2 diabetes -Global burden of disease and forecasted trends. J Epidemiol Glob Health [Internet]. 2020 [cited 2021 January 29];10(1):107–11. Available from: /pmc/articles/PMC7310804/?report=ab stract
- Cho NH, Shaw JE, Karuranga S, Huang Y, da Rocha Fernandes JD, Ohlrogge AW, et al. IDF Diabetes Atlas: Global estimates of diabetes prevalence for 2017 and projections for 2045. Diabetes Res Clin Pract [Internet]. 2018 [cited 2021 January 29];138:271– 81. Available from: https://doi.org/ 10.1016/j.diabres.2018.02.023
- Baena-Díez JM, Peñafiel J, Subirana I, Ramos R, Elosua R, Marín-Ibañez A, et al. Risk of cause-specific death in individuals with diabetes: A competing risks analysis. Diabetes Care [Internet]. 2016 [cited 2021 January 29];39(11): 1987–95. Available from: http://www. diabetesjournals
- American Diabetes Association. Standard medical care in diabetes 2018. J Clin Appl Res Educ 2018.
- Chatterjee S, Davies MJ, Heller S, Speight J, Snoek FJ, Khunti K. Diabetes structured self-management education programmes: a narrative review and current innovations. Lancet Diabetes Endocrinol. 2017;6(2):130– 42.
- Haas L, Maryniuk M, Beck J, Cox CE, Duker P, Edwards L, et al. National standards for diabetes self-management education and support. Diabetes Educ. 2014;38:5.
- Beck J, Greenwood DA, Blanton L, Bollinger ST, Butcher MK, Condon JE, et al. 2017 National Standards for Diabetes Self-Management Education

and Support. Diabetes Care. 2017; 40(10):1409–19.

- 10. Armstrong C. ADA Updates Standards of Medical Care for Patients with Diabetes Mellitus [Internet]. American Family Physician 2017 [cited 2021 January 27]. Available from: www.aafp.org/afp/
- 11. Chai S, Yao B, Xu L, Wang D, Sun J, Yuan N, et al. The effect of diabetes self-management education on psychological status and blood glucose in newly diagnosed patients with diabetes type 2. Patient Educ Couns [Internet]. 2018;101(8):1427–32. Available from: https://doi.org/ 10.1016/j.pec.2018.03.020
- 12. Adam L, O'Connor C, Garcia AC. Evaluating the Impact of Diabetes Self-Management Education Methods on Knowledge, Attitudes and Behaviours of Adult Patients With Type 2 Diabetes Mellitus [Internet]. Can J Diabetes. 2018 [cited 2020 October 14];42(5):470-477.e2. Available from: https://doi.org/10.1016/j.jcjd.2017.11.0 03
- Abdulah DM, Hassan AB, Saadi FS, Mohammed AH. Impacts of selfmanagement education on glycaemic control in patients with type 2 diabetes mellitus [Internet]. Diabetes Metab Syndr Clin Res Rev. 2018;12(6):969– 75. Available from: http://www. sciencedirect.com/science/article/pii/S 1871402118302145
- 14. Orbell S, Schneider H, Esbitt S, Gonzalez JS, Gonzalez JS, Shreck E, et al. Hemoglobin A1c. In: Encyclopedia of Behavioral Medicine [Internet]. Springer New York; 2013 [cited 2021 January 29]. p. 957–957. Available from: https://www.ncbi.nlm.nih.gov/ books/NBK549816/
- 15. Mohebi S, Rad G, Bakht L, Feizi A. Importance of social support in diabetes care. J Educ Health Promot [Internet]. 2013 [cited 2021 Jan 29];2(1):62.

Available from: /pmc/articles/ PMC3908488/?report=abstract

- 16. Mayberry LS, Osborn CY. Family support, medication adherence, and glycemic control among adults with type 2 diabetes. Diabetes Care [Internet]. 2012 Jun [cited 2021 Feb 3];35(6):1239–45. Available from: https://pubmed.ncbi.nlm.nih.gov/2253 8012/
- 17. Pamungkas RA, Chamroonsawasdi K, Vatanasomboon P. A systematic review: Family support integrated with diabetes self-management among uncontrolled type II diabetes mellitus patients [Internet]. Behav Sci (Basel); 2017 [cited 2020 Oct 14];7(3):1–17. Available from: www.mdpi.com/ journal/behavsci
- Moher D. Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. Ann Intern Med [Internet]. 2009 [cited 2020 December 7];151(4):264. Available from: http://annals.org/ article.aspx?doi=10.7326/0003-4819-151-4-200908180-00135
- 19. Higgins JPT, Altman DG, Gøtzsche PC, Jüni P, Moher D, Oxman AD, et al. The Cochrane Collaboration's tool for assessing risk of bias in randomised trials [Internet]. BMJ; 2011 [cited 2020 December 7];343(7829). Available from: http://www.bmj.com/content/ 343/bmj.d5928/suppl/DC1
- 20. Powers MA, Bardsley J, Cypress M, Duker P, Funnell MM, Fischl AH, et al. Diabetes self-management education and support in type 2 diabetes: A joint position statement of the American Diabetes Association, the American Association of Diabetes Educators, and the Academy of Nutrition and Dietetics [Internet]. Diabetes Care; 2015 [cited 2020 Dec 7];38(7):1372–82. Available from: http://www.diabeteseducator. org/

- 21. Guyatt GH, Oxman AD, Vist GE, Kunz R, Falck-Ytter Y, Alonso-Coello P, et al. GRADE: An emerging consensus on rating quality of evidence and strength of recommendations [Internet]. BMJ; 2008 [cited 2020 December 7]; 336(7650):924–6. Available from: https://www.bmj.com/content/336/765 0/924
- 22. Godavitarne C, Robertson A, Ricketts DM, Rogers BA. Understanding and interpreting funnel plots for the clinician [Internet]. Br J Hosp Med; 2018 [cited 2020 December 19];79(10):578–83. Available from: https://pubmed.ncbi.nlm.nih.gov/3029 0744/
- 23. Begg CB, Mazumdar M. Operating Characteristics of a Rank Correlation Test for Publication Bias. Biometrics. 1994;50(4):1088.
- 24. Egger M, Smith GD, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test [Internet]. Br Med J; 1997 [cited 2021 March 18];315(7109):629–34. Available from: https://www.bmj.com/ content/315/7109/629
- 25. Higgins JPT, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses [Internet]. Vol. 327, British Medical Journal. BMJ Publishing Group; 2003 [cited 2020 December 7]. p. 557–60. Available from: /pmc/articles/PMC192859/? report=abstract
- 26. Riley RD, Higgins JPT, Deeks JJ. Interpretation of random effects metaanalyses [Internet]. BMJ. 2011 [cited 2020 December 7];342(7804):964–7. Available from: https://www.bmj.com/ content/342/bmj.d549
- 27. Lewis S, Clarke M. Forest plots: Trying to see the wood and the trees [Internet].
 Vol. 322, British Medical Journal: BMJ Publishing Group; 2001 [cited 2020 December 7]p. 1479–80. Available

from: https://www.ncbi.nlm.nih.gov/ pmc/articles/PMC1120528/

- 28. Brown SA, Garcia AA, Kouzekanani K, Hanis CL. Culturally competent diabetes self-management education for Mexican Americans: The Starr County border health initiative. Diabetes Care. 2002;25(2):259–68.
- 29. Vincent D, Pasvogel A, Barrera L. A feasibility study of a culturally tailored diabetes intervention for mexican americans. Biol Res Nurs. 2007;9(2):130–41.
- 30. Rosal MC, Ockene IS, Restrepo A, White MJ, Borg A, Olendzki B, et al. Randomized trial of a literacysensitive, culturally tailored diabetes self-management intervention for lowincome Latinos: Latinos en control. Diabetes Care. 2011;34(4):838–44.
- 31. Toobert DJ, Strycker LA, Barrera M, Osuna D, King DK, Glasgow RE. Outcomes from a multiple risk factor diabetes self-management trial for latinas: ¡viva Bien! Ann Behav Med. 2011;41(3):310–23.
- 32. Castejón AM, Calderón JL, Perez A, Millar C, McLaughlin-Middlekauf J, Sangasubana N, et al. A community -Based pilot study of a diabetes pharmacist intervention in latinos: Impact on weight and hemoglobin A1c. J Health Care Poor Underserved 2013;24(4 SUPPL.):48–60.
- 33. García AA, Brown SA, Horner SD, Zuñiga J, Arheart KL. Home-based diabetes symptom self-management education for Mexican Americans with type 2 diabetes. Health Educ Res. 2014;30(3):484–96.
- 34. Trief PM, Fisher L, Sandberg J, Cibula DA, Dimmock J, Hessler DM, et al. Health and psychosocial outcomes of a telephonic couples behavior change intervention in patients with poorly controlled type 2 diabetes: A randomized clinical trial. Diabetes Care

[Internet]. 2016 [cited 2020 October 14];39(12):2165–73. Available from: http://care.diabetesjournals.org/lookup/ doi/10.2337/dc16-0035

- 35. Ing CT, Zhang G, Dillard A, Yoshimura SR, Hughes C, Palakiko D-MM, et al. Social Support Groups in the Maintenance of Glycemic Control after Community-Based Intervention. J Diabetes Res [Internet]. 2016 [cited 2020 October 14];2016:1–8. Available from: http://www.hindawi.com/ journals/jdr/2016/7913258/
- 36. McEwen MM, Pasvogel A, Murdaugh C, Hepworth J. Effects of a Family-Intervention based Diabetes on Behavioral and Biological Outcomes for Mexican American Adults. Diabetes Educ [Internet]. 2017 [cited 2020 October 14];43(3):272-85. http://journals. Available from: sagepub.com/doi/10.1177/0145721717 706031
- 37. McElfish PA, Long CR, Kohler PO, Yeary KHK, Bursac Z, Narcisse MR, et al. Comparative effectiveness and maintenance of diabetes selfmanagement education interventions for Marshallese patients with type 2 diabetes: A randomized controlled trial. Diabetes Care. 2019;1:42(5):849–58.
- Shakibazadeh E, ... LB-H promotion, 2015 undefined. Persian diabetes selfmanagement education (PDSME) program: evaluation of effectiveness in Iran. academic.oup.com [Internet]. [cited 2020 Oct 14]. Available from: https://academic.oup.com/heapro/articl e-abstract/31/3/623/1749576
- 39. Tabasi HK, Madarshahian F, Nikoo MK, Hassanabadi M, Mahmoudirad G. Impact of family support improvement behaviors on anti diabetic medication adherence and cognition in type 2 diabetic patients. J Diabetes Metab Disord 2014;13(1):1–6.
- 40. Wichit N, Mnatzaganian G, Courtney M, Schulz P, Johnson M. Randomized

controlled trial of a family-oriented self-management program to improve self-efficacy, glycemic control and quality of life among Thai individuals with Type 2 diabetes. Diabetes Res Clin Pract [Internet]. 2017 [cited 2020 October 14];123:37–48. Available from: http://dx.doi.org/10.1016/ j.diabres.2016.11.013

- 41. Withidpanyawong U, Lerkiatbundit S, Saengcharoen W. Family-based intervention by pharmacists for type 2 diabetes: A randomised controlled trial. Patient Educ Couns [Internet]. 2019 [cited 2020 October 14];102(1):85–92. Available from: https://linkinghub. elsevier.com/retrieve/pii/S0738399118 305354
- 42. Keogh KM, Smith SM, White P, McGilloway S, Kelly A, Gibney J, et al. Psychological family intervention for poorly controlled type 2 diabetes. Am J Manag Care. 2011;17(2):105–13.
- 43. Kang CM, Chang SC, Chen PL, Liu PF, Liu WC, Chang CC, et al. Comparison of family partnership intervention care vs. conventional care in adult patients with poorly controlled type 2 diabetes in a community hospital: A randomized controlled trial. Int J Nurs Stud [Internet]. 2010;47(11):1363–73. Available from: http://dx.doi.org/ 10.1016/j.ijnurstu.2010.03.009
- 44. Gomes LC, Coelho ACM, Gomides D dos S, Foss-Freitas MC, Foss MC, Pace AE. Contribution of family social support to the metabolic control of people with diabetes mellitus: A randomized controlled clinical trial. Appl Nurs Res. 2017;1:36:68–76.
- 45. Trief PM, Fisher L, Sandberg J, Hessler DM, Cibula DA, Weinstock RS. Two for one? Effects of a couples intervention on partners of persons with Type 2 diabetes: a randomized controlled trial. Diabet Med [Internet]. 2019 [cited 2020 October 14]; 36(4):473–81. Available from: http://

doi.wiley.com/10.1111/dme.13871

- 46. Azami G, Soh KL, Sazlina S-G, Salmiah MS, Aazami S. Behavioral interventions to improve selfmanagement in Iranian adults with type 2 diabetes: a systematic review and meta-analysis. J Diabetes Metab Disord [Internet]. 2018 [cited 2020 October 14];17(2):365–80. Available from: http://link.springer.com/10.1007/s4020 0-018-0376-0
- 47. Ramírez-Mendoza F, González JE, Gasca E, Camacho M, Cruz M V., Caraveo D, et al. Time in range and HbA1C after 6 months with a multidisciplinary program for children and adolescents with diabetes mellitus, real world data from Mexico City. Pediatr Diabetes [Internet]. 2020 [cited 2020 December 19];21(1):61–8. Available from: /pmc/articles/ PMC6973224/?report=abstract
- 48. Steinsbekk A, Rygg L, Lisulo M, Rise MB, Fretheim A. Group based diabetes self-management education compared to routine treatment for people with type 2 diabetes mellitus. A systematic review with meta-analysis. Vol. 12, BMC Health Services Research. 2012.
- 49. Stratton IM, Adler AI, Neil HAW, Matthews DR, Manley SE, Cull CA, et al. Association of glycaemia with and microvascular macrovascular complications of type 2 diabetes (UKPDS 35): Prospective study. observational Br Med J [Internet]. 2000 [cited 2021 March 23];321(7258):405-12. Available from: /pmc/articles/PMC27454/
- 50. Ikeda M, Shimazawa R. Challenges to hemoglobin A1c as a therapeutic target for type 2 diabetes mellitus [Internet]. Vol. 20, Journal of General and Family Medicine. Wiley-Blackwell; 2019 [cited 2021 Febuary 1]. p.129–38. Available from: /pmc/articles/ PMC6612881/?report=abstract

- 51. Lage MJ, Boye KS. The relationship between HbA1c reduction and healthcare costs among patients with type 2 diabetes: evidence from a U.S. claims database. Curr Med Res Opin [Internet]. 2020 [cited 2021 February 1];36(9):1–7. Available from: https:// pubmed.ncbi.nlm.nih.gov/32643451/
- 52. Wilding J, Godec T, Khunti K, Pocock S, Fox R, Smeeth L, et al. Changes in HbA1c and weight, and treatment persistence, over the 18 months following initiation of second-line therapy in patients with type 2 diabetes: results from the United Kingdom Clinical Practice Research Datalink. BMC Med [Internet]. 2018 [cited 2021 February 1];16(1):116. Available from: https://bmcmedicine.biomedcentral.co m/articles/10.1186/s12916-018-1085-8
- 53. Gary TL, Genkinger JM, Guallar E, Peyrot M, FL. B. Meta-analysis of randomized educational and behavioral interventions in type 2 diabetes. Diabetes Educ. 2003;29:3.
- 54. Riddle MC, Vlajnic A, Zhou R, Rosenstock J. Baseline HbA1c predicts attainment of 7.0% HbA1c target with structured titration of insulin glargine in type 2 diabetes: A patient-level analysis of 12 studies. Diabetes, Obes Metab [Internet]. 2013 [cited 2021 February 1];15(9):819–25. Available from: https://pubmed.ncbi.nlm.nih.gov/2348 9438/
- 55. Association AD. Standards of medical care in diabetes 2013 [Internet]. Vol. 36, Diabetes Care. American Diabetes Association; 2013 [cited 2021 February 2]p.S11. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC35 37269/
- 56. Carpenter R, DiChiacchio T, Barker K. Interventions for self-management of type 2 diabetes: An integrative review [Internet]. Vol. 6, International Journal of Nursing Sciences. Chinese Nursing

Association; 2019 [cited 2021 February 1]p.70–91. Available from: /pmc/ articles/PMC6608673/?report=abstract

- 57. Chrvala CA, Sherr D, Lipman RD. Diabetes self-management education for adults with type 2 diabetes mellitus: A systematic review of the effect on glycemic control. Patient Educ Couns. 2016;1:99(6):926–43.
- 58. He X, Li J, Wang B, Yao Q, Li L, Song R, et al. Diabetes self-management education reduces risk of all-cause mortality in type 2 diabetes patients: a systematic review and meta-analysis. Endocrine [Internet]. 2017 [cited 2020 October 14];55(3):712–31. Available from: http://link.springer.com/10.1007/ s12020-016-1168-2
- 59. Cunningham AT, Crittendon DR, White N, Mills GD, Diaz V, Lanoue MD. The effect of diabetes selfmanagement education on HbA1c and quality of life in African-Americans: A systematic review and meta-analysis. BMC Health Serv Res. 2018;18(1).
- 60. Qi L, Liu Q, Qi X, Wu N, Tang W, Xiong H. Effectiveness of peer support for improving glycaemic control in patients with type 2 diabetes: A metaanalysis of randomized controlled trials. BMC Public Health. 2015;15(1).
- 61. Winkley K, Upsher R, Stahl D, Pollard D, Kasera A, Brennan A, et al. Psychological interventions to improve self-management of type 1 and type 2 diabetes: A systematic review. Health Technol Assess (Rockv) [Internet]. 2020 [cited 2021 March 21];24(28):1–268. Available from: https://www.ncbi.nlm.nih.gov/books/NBK558398/